

Banc Ceannais na hÉireann Central Bank of Ireland

Eurosystem

Research Technical Paper

Wealth accumulation and inter-generational inequality with inverted population pyramids

Simone Cima Vol. 2022, No. 6 August 2022

Wealth accumulation and inter-generational inequality with inverted population pyramids

Simone Cima*

Abstract

Demographic dynamics and the shift of population pyramids towards an invertedpyramid shape in advanced economies are leading to relative scarcity of labour and excess savings. What are the effects of these dynamics on the relative wealth accumulation journeys of different cohorts? Within a fixed-effect crosscountry panel framework, I find that savings by an increasing share of households aged between 45 and 65, a rise in retired over-65s, and a decrease in working-age and low-wealth agents in their twenties and thirties can explain most of the decline in rates of return across countries in the last few decades, and similarly a large part of the increase in wages. In this context and looking to the future, wealth accumulation out of income and capital returns by cohorts living in advanced economies and retiring in future decades is set to become increasingly difficult, as higher wages are not sufficient to compensate for lower returns over long periods of time. Current young and future generations are therefore set to face progressively lower standards of living at retirement and/or increasingly high saving ratios in working age.

KEYWORDS: Demographics, Rates of return, Inter-generational inequality, Wealth

JEL Classification: E21, E25, J11.

^{*}Simone Cima, Central Bank of Ireland, contact: simone.cima@centralbank.ie. My thanks go to Joseph Kopecky, Stephen Byrne, Giuseppe Corbisiero, Robert Kelly, Reamonn Lydon, Gerard O'Reilly, Elena Sponga, Michael Wycherley, staff and fellow PhD students at Trinity College Dublin, and colleagues in Irish Economic Analysis and Monetary Policy at the Central Bank of Ireland for helpful comments. Views expressed in this paper are my own and do not necessarily reflect those of the Central Bank of Ireland, the European Central Bank, or the ESCB. All errors are my own.

Non-technical summary

Over the past decades, as a result of strong increases in life expectancies and declines in birth rates, populations in economies across the world have been going through a demographic transition, with falling population growth rates, and a general ageing of the population. These dynamics are at the most advanced stage in a number of developed economies: in these economies, population pyramids have started taking the shape of *inverted* pyramids, with large shares of the population at older ages and progressively smaller younger age groups. Such population structures are unprecedented.

Because households of different ages exhibit different patterns of consumption and saving, as well as different participation rates in the labour force, any change in the relative weight of a certain age group compared to the overall population will have an aggregate effect on the economy's consumption, saving, and the size of the labour force. The demographic trends we are witnessing in most advanced economies are shifting the weight of the population towards the age groups with the highest levels of accumulated savings (i.e. households in late working age and early retirement), while at the same time the share of individuals in late retirement is on the increase and the share of the population in early working age is falling, lowering the potential size of the labour force.

By differently affecting the employment rate of the population as well as aggregate wealth levels, each age group has a different impact on rates of return on capital and on wages, as these are affected by the quantity of labour and capital in the economy. In general, this paper finds that population structures of the type present in ageing economies are associated with more capital and less labour, typically leading to lower returns on capital and higher wages. As economies transition from a young (high-return, lower wages) population structure towards inverted population pyramids (low returns, higher wages), the ability of cohorts born at different times through the transition to accumulate wealth is affected.

While younger cohorts can benefit from higher wages thanks to both technological advancement and the effects of ageing, I find that the negative impact of lower returns on wealth accumulated over a long period of time tends to prevail (conditional on assumptions), leading to these cohorts not being able to accumulate as much wealth as those preceding them from the same saving rates. Because younger cohorts will also need to accumulate more wealth than the previous cohorts on account of longer life expectancies after retirement, they potentially face a trade-off between decreasing standards of living at retirement and higher saving (i.e. lower consumption) during their working age.

1 Introduction

The world is in the midst of one of the most dramatic demographic shifts in history. After two centuries of exponential expansion, world population growth reached its peak in the 70s and has experienced a considerable slowdown since. The ageing of societies is already at an advanced stage in many developed economies, some of which face population declines. This is quickly becoming a global phenomenon. Life expectancies have never been as high; meanwhile, birth rates are well below replacement in an increasing number of economies.

Population pyramids in a number of countries have now started taking the shape of *inverted* pyramids, with younger generations progressively smaller and a high median age - see Figure 1 for the striking example of Japan. Population structures of this type are unprecedented. An ageing and eventually declining population has profound consequences for the prospects of an economy, in terms of productivity, equality, allocation of resources, financial markets and economic policy.



Figure 1: Evolution of the Japanese population pyramid.

Percentage of the population in each 5-year age group.

As longer lifespans push larger shares of workers towards retirement age, while the relative size of the working population shrinks, an obvious question arises: how to finance the livelihoods of current retirees while allowing the working-age population to accumulate enough wealth to fund their own retirement? With public finances increasingly strained by pension payments, a large part of retirees' income might need to originate from the private wealth accumulated during their working life. The accumulation of sufficient amounts of wealth requires not only a steady stream of savings, but also high enough real rates of return on such savings. However, population aging itself might have an adverse effect on rates of return on wealth; empirical results obtained in this paper give further evidence of this.

In the context of demographic ageing and decline, younger generations in an increasing number of countries have been worrying that their future economic prospects look rather bleak when compared to their parents' generations. When their parents could, for instance, afford a house and start a family early in life, this is becoming increasingly difficult for many young people in advanced economies, who are facing a situation of low savings with low returns, and high asset prices. Are these worries founded, and do demographic dynamics have anything to do with trends in wealth inequality?

The main purpose of this paper is to explore and quantify how the wealth accumulation journey of different generations (or cohorts) has been or will be affected by demographic trends, and to derive implications in terms of inequality (of wealth, but more generally, of standards of living) between these cohorts. While the analysis within the paper explores these dynamics across the world, particular emphasis is given to older, advanced economies at the frontier of the demographic transition.

In these economies, how do young workers today fare compared to their parents' generation? Generally speaking, will they be worse-off? I find that, in advanced, ageing economies, that appears indeed to be the case. Consistent with a growing literature, this paper presents evidence that ageing pushes down rates of return, but also positively affects wages. As these dynamics unfold, accumulating wealth becomes relatively more difficult for at least some section of the population. Because returns will be lower, any initial amount of saving will not accumulate as quickly. At the same time, wages will increase, but not enough for households to be able to reach the same levels of wealth accumulation as the generations before them with the same saving rates, unless unlikely high productivity growth is assumed in future decades. Younger generations will thus have to save more to reach the same levels of wealth as their parents, and even more to reach a level of wealth that will allow them to have an equally comfortable retirement (seeing as life expectancies are set to increase further).

I reach these findings by employing a simple fixed-effect panel estimation approach. I find that ageing demographics tend to depress rates of return and increase wages, although with different intensities, through the effect of the population structure on capital accumulation in the economy and employment. The estimation method used also allows me to show how each different age group has a markedly different impact on returns and wages through these two channels. By combining the single age-group effects together, I can then

recover how the whole population pyramid affects rates of return and wages, and wealth accumulation as a result, throughout countries and years. With a number of quantitative exercises and back of the envelope calculations, I then finally quantify the implications of these dynamics for the wealth accumulation journeys of different cohorts, and how they fare compared to one another, reaching the conclusion that current young generations and those in the future are set to lose, especially compared to the generation currently at retirement age, which in the average advanced economy benefited from the most favourable combination of demographic and economic conditions.

The rest of the paper is structured as follows: Section 2 reviews some of the literature on demographics, wealth accumulation, the macroeconomy and inequality, and locates this paper in the context of this literature; Section 3 introduces the data used in the paper and addresses some issues related to these data; Section 4 looks at the impact of demographic dynamics on rates of return through different channels and explores the implications for wealth accumulation and inter-generational inequality; Section 5 repeats this analysis for wages; Section 6 combines these findings to obtain the overall impact of demographics on wealth accumulation across generations; finally, Section 7 concludes the paper.

2 Literature Review

This paper draws from a vast and established literature on demographics and the economy, of which I can identify a few strands that are most relevant in particular to the effects of demographic dynamics on rates of return, and inequality.

One of the main channels through which demographic developments affect the economy is through savings. The literature on life-cycle consumption and savings generally finds that the consumption and saving behaviour of economic agents changes as they age, as their earnings and levels of wealth change, and they prepare for retirement. Households start their working life with low earnings and quickly increase their incomes, and earn most close but before retirement age, after which their income falls. Consumption is also humpshaped and partly follows the evolution of income, however saving rises by more for several decades before declining sharply and becoming negative well into retirement. The literature on life-cycle behaviour is vast and established - see for instance Gourinchas and Parker (2002) (17) for a life-cycle representation of income and consumption. Results in Feiveson and Sabelhaus (2019) (13) show that annual saving is typically highest when an agent is between 35 and 50, while wealth increases at the fastest pace between 40 and 60 (due to effects of returns on accumulated wealth). Aggregating this type of saving and consumption behaviour in the macroeconomy results in large effects of age on aggregate wealth, consumption, and saving rates. In particular, a large share of the population in the ages with the highest accumulated lifetime wealth will lead to higher average levels of wealth

per person in the economy, with implications for interest rates.

There is ample evidence, both theoretical and empirical, that population ageing leads to a reduction in interest rates - whether by interest rate we intend the natural interest rate or rates of return or other measures - although there is disagreement on the magnitude and on how other effects are also at play. A large number of papers reach these conclusions by employing an overlapping-generations (OLG) model; these for instance include Geppert, Ludwig and Abiry (2016) (15), Börsch-Supan et al. (2003) (4), and Papetti (2019) (27); Krueger and Ludwig (2006) (24) also find that, in an open economy, capital flows cause economies to "export" or "import" part of the decline in interest rates due to demographics (respectively, if ageing happens more or less rapidly compared to ROW). Carvalho, Ferrero and Nechio (2016) (5) again find, within a life-cycle model, that ageing demographics lead to a reduction in interest rates, via increasing capital per worker and higher life expectancies, although lower savings by retirees have an opposite but insufficient effect (similar results are found in Papetti (2019) (27) and Börsch-Supan et al. (2003) (4)). Ikeda and Saito (2012) (20) attribute a greater part of the reduction in interest rates in Japan to a decline in TFP, but demographics also have a large negative effect. That ageing demographics put downward pressure on interest rates is also documented empirically - see for instance Ferrero, Gross and Neri (2017) (11).

A reduction in interest rates, generally, tends to be a collateral finding of a great part of theoretical papers with demographic dynamics. These multiple results are all consistent with the literature on the life-cycle patterns of savings and consumption, as in general a higher proportion of the population in ages with high accumulated savings leads to an excess of these relative to investment, thus lowering rates. Some papers however challenge the view that demographics are the main factor responsible for decreasing rates. Goodhart and Pradhan (2017) (16) mostly attribute falling interest rates in the past decades to a shift of manufacturing production to emerging markets (especially China), and rather see ageing demographics globally as potentially reversing the trend in the near future. Mian, Straub and Sufi (2021) (25) mostly associate the fall in the natural rate of interest in the US to increases in income inequality leading to higher saving by the wealthier.

In this paper, I find again that ageing demographic structures lead to lower rates of return on capital through higher capital accumulation and lower employment. Interestingly however, thanks to the estimation I employ, I am also able to go further and break down the effects on rates of return by age group and to quantify how each specific shape of the population pyramid might impact returns. The results suggest that demographic dynamics may account for most of the variation in rates of return across the years in the countries studied.

Demographic ageing is not only associated with lower interest rates, but also with lower potential growth - papers that make this argument include Cooley and Henriksen (2017) (7) and Aksoy et al. (2019) (1). Within a context of population ageing and low or negative population growth, combined with anemic productivity growth and low interest rates (which due to demographic dynamics are all seen occurring together and persisting for the rest of the century) part of the literature talks about secular stagnation - see for instance the already cited Geppert, Ludwig and Abiry (2016) (15), and Ferrero, Gross and Neri (2017) (11). Demographics are only one of the potential explanations for the low interest rate environment in advanced economies and secular stagnation - both Sajedi and Twaites (2016) (28), and Eichengreen (2015) (9), for instance, make the case for the fall in the relative price of investment goods being one contributing factor. In his famous speech, Summers (2014) (29) lists six structural reasons for the decline in natural interest rates (which he links to secular stagnation) experienced in the US, and out of these, one is the slowdown in population growth. While in this paper I do not make considerations on economic growth and productivity as it relates to demographics, this strand of the literature informs the analysis and interpretation of results.

With life-cycle savings, rates of return and employment as main channels, the literature tends to find that ageing demographics lead to increasing wealth inequality, particularly of the inter-generational type, although there is no general consensus. Antunes and Ercolani (2020) (3) find that demographic dynamics can explain a large part of the increase in intergenerational wealth inequality in the US. Geppert (2015) (14) finds that demographic effects on wages, human capital accumulation and the skill premium will lead to different welfare effects across household types. Vandenbroucke (2016) (31) sees increases in life expectancy as increasing wealth inequality while however declines in population growth decrease it. Differently from most of this literature, this paper does not look at inter-generational wealth inequality from a cross-sectional perspective (how wealth is distributed among households of different ages in the same year), but rather examines inequality between generations in terms of cohorts, i.e. how agents born in different years compare to one another at the same age.

I focus on the different economic environment faced by different cohorts, and analyse wealth inequality between generations under this light. In Hood and Joyce (2013) (19), the authors show how in spite of higher real incomes at comparable ages, younger cohorts in the UK are so far not keeping up with previous generations in building up wealth or accessing housing. The principal contribution of this paper is to reach similar findings within a cross-country empirical framework, and additionally linking demographics as a contributing factor to these developments; I also show plausible projections for wealth accumulation for a large number of countries up to the end of the century. In addition, the results of this paper provide a specific age-bracket dimension to the well-documented negative effect of ageing demographics on rates of return and positive wage effect.

3 Data

3.1 Sources

The analysis performed in this paper makes use of a large panel dataset. Data on macroeconomic aggregates such as GDP and other national accounts components, the capital stock, employment etc. are taken from version 9.1 of the Penn World Tables (PWT 9.1) (12). The Penn World Tables are also the source of data on productivity, capital share of income and depreciation, among others. PWT variables are used to compute the return on capital, appreciation of capital, and its aggregate rate of return as detailed below in the following subsection. The dataset includes 182 countries between 1950 and 2017.

Data on countries' population are gathered from the United Nations' World Population Prospects 2019 (30). These include projections on the population of countries at each year from 1950 up to 2100 and are broken down by five-year age groups. This allows for a detailed analysis of the evolution of the demographic structure of a country over time, both in terms of growth and of age distribution.

I combine the PWT 9.1 and UN population prospects data to create a panel dataset of 64 countries, spanning between the years 1950 and 2100, where the Penn World Tables data stops at year 2017. The countries were chosen based on having a sufficient number of time series observations for the most relevant variables to the analysis conducted in this paper; having typically better data, most of the advanced economies are respresented in this final sample, with however around 20 developing economies (depending on definitions) represented as well. The final panel used for regression analysis in Section 4 contains about 3,000 observations, or an average of almost 47 years for each of the 64 countries in the panel.

The economies included in the panel are crucially at very different stages of aging and population growth, and have quite different population distributions, life expectancies and birth rates. For the countries with the most complete data, this can go back as far as 1950 and up to 2017, allowing to cover a time span of almost seventy years, during which many economies, especially in the developed world, experienced radical shifts in their population dynamics, from growth, birth rates, age distribution and migration patterns. During these years, life expectancy also quickly increased all across the world.

Before proceeding to the empirical analysis of these data, the next two subsections explore a few data issues more in detail.

3.2 Total Return on Capital

This paper investigates how demographics affect wealth inequality through the lens of rates of return. Computing an exact measure of real return on wealth requires time series of price indices as well as yields (i.e. dividends, interest, rents, etc.) for all asset classes, and the relative weight of each asset class in a typical household's portfolio. This is not generally possible apart from a handful of countries, and some strong assumptions.

The alternative approach used here to estimate return on wealth draws from David, Henriksen and Simonovska (2014) (8) and makes use of macroeconomic data. I compute returns on an economy's capital by considering what would be the before-tax annual gains, realised and unrealised, accruing to a representative agent investing into the capital of such economy, as a fraction of the amount of capital initially invested. Here I consider the returns accruing to a domestic investor who invests into the "average" unit of domestic capital.¹

While wealth can take the form of various types of financial and non-financial assets, aggregating returns across asset types allows me to treat aggregate wealth, and its return, as equivalent to a generic capital good. In this way, developments in the return to capital can be equated to returns to the portfolio of a household with the average wealth allocation. Ultimately, returns on each type of asset can be associated with returns generated with the use of physical capital somewhere, with housing assets being the most straightforward. The return on government bonds, for instance, is linked to the domestic capital held by the government, as well as the collection of taxes that are again collected from income generated from the use of private domestic capital.

Treating domestic agents' wealth and an economy's capital as equivalent crucially relies on the assumption of closed capital markets, where all the wealth of domestic agents is invested into domestic capital, and there is no capital flow. Any differential between the return on domestic assets, and the return on the wealth of domestic agents, is due to the openness of capital and financial markets, and the extent to which agents own higher- or lower-yielding foreign capital in their portfolios.² While capital flows may substantially affect this differential, there is very strong evidence of home bias in equity markets (see for instance Cooper et al. (2013)(6))³ and in government bond markets, while the largest component of wealth in most households' portfolios (and largest component of any economy's capital stock), i.e. residential housing, is in very large part held by domestic agents. Indeed, the wealth owned by the median household in most countries tends to be even more homebiased than the average, as typical households have very little exposure to equity markets and hold most of their wealth in domestic housing. For the above reasons, the rest of this

¹This also removes complications related to changes in relative prices of consumption goods between the investor's country and the country where capital is invested.

²Recall the findings in Krueger and Ludwig's (2006) (24) model.

³Compare also Mishra (2015) (26), who finds an average equity home bias of 0.77 (with 1 being complete home bias) across 42 countries.

paper treats returns on an economy's capital and on its citizens' wealth as essentially equivalent. $^{4}\,$

A measure of total return on domestic capital can be computed as the sum of two components:

- Income flows accruing to capital, such as payment of interest, dividends, or rents, depending on the type of asset, but also profits retained by corporations. These are equivalent to total income accruing to capital, divided by the stock of capital, and can be retrieved from the data as $Y_t^K = \alpha Y_t/K_t$, with α being the capital share of income. This can be interpreted as the marginal product of capital.
- Real changes in the price of capital, relative to consumption goods (appreciation of capital), $(P_{t+1}^K P_t^K)/P_t^K$, where P_t^K represents the real price of capital goods relative to consumption goods, at time t.

Total return on capital at time t (TRK_t) is thus:

$$TRK_{t} = \alpha \frac{Y_{t}}{K_{t}} + \frac{P_{t+1}^{K} - P_{t}^{K}}{P_{t}^{K}}$$
(1)

with all the necessary data available within the Penn World Tables.

While it is informative to think of returns on capital in terms of these two components, it turns out that what matters for the questions posed by this paper is exclusively the first component. While Y^K in the long term depends on productive capacity considerations alone, which are slow moving, persistent and closely interconnected with demographic developments, the appreciation of capital appears to be a short-term phenomenon more related to the business cycle.

Appreciation of capital averages at almost exactly 0% per year for the countries in the panel, signifying that there is no general evidence of capital goods increasing in value relative to consumption goods. However, it is interesting to note that while capital has not appreciated significantly across the world in the last 70 years, this net-zero appreciation hides the fact that prices of one component of it, residential and non-residential structures, have been steadily rising while the other types of capital (machinery, transport equipment, and other) have become relatively cheaper over the years.

To illustrate this, Figure 2 shows how the average price level of the aggregate capital stock in real terms has broadly remained unchanged, while there has been a clear trend upwards

⁴Even when domestic agents invest in other economies, they tend to be biased towards countries with similar characteristics to their own, including in terms of demographic dynamics. To the extent to which demographic factors influence rates of return, as this paper argues, investing in a country with similar demographics is almost equivalent to investing domestically, further supporting this argument.



Figure 2: Average Real Capital Price Index, and Components

Real Capital Price Index (RKPI), for aggregate capital, machinery and equipment, transport equipment, and residential and non-residential structures (other assets omitted); unweighted average values across the countries in the panel (2011 = 1). For each country, the indices are calculated as the capital stock price index (PWT, 2011 = 1) divided by the GDP deflator (2011 = 1).

in the prices of buildings, and downwards in the other types of capital. Although the following section will be devoted to the analysis of developments in returns to aggregate capital only, this fact has potentially very important consequences for developments in wealth inequalities and should be kept in mind. The potential effects of such an appreciation of the value of buildings (and especially housing, which represents the largest component of most households' portfolios) on wealth inequality are discussed briefly later, however further research is warranted to fully evaluate the extent to which demographic developments have anything to do with these trends.

3.3 Population and Demographic Distribution

Populations change extremely slowly, which makes estimating the effects of such changes on economic variables, which typically have considerably more volatility, particularly challenging. Demographic changes are felt and experienced only over multiple years and even decades, as opposed to months and quarters. Nevertheless, demography remains at all times one of the most fundamental factors underpinning the performance and direction of an economy.

For this reason, while births, deaths and migration are what ultimately drive populations, because their effect is only seen years later, this paper mainly looks at demographics through the lens of the age-structure of an economy's population, or the population pyramid. Population pyramids are the result of a combination of past birth rates and death rates, evolutions in life expectancy and migration, but adequately represent where an economy's population is at the moment and where it is going. I will make use of these for most of my analysis below.

The data from the UN Population Prospects include countries' populations by five-year age groups, by number of persons and as a share of the total population in the country. This results in 18 5-year age groups (85+ year-olds are here aggregated into one only group due to the small size of the population in this age bracket), starting from 0 to 4 year-olds, and so on.

I take the working-age population to be the population between the ages of 20 and 64, inclusive. This appears to be the most appropriate choice when taking account of crosscountry differences (and even differences within a country throughout the period under analysis) regarding pension age and typical age at which people get their first job.

Regression analysis presented below seeks to determine the effect of the relative size of each age-group population share on a number of variables. To do so, the first empirical approach could be to add each of these shares into a regression. However, population shares are collinear, and their sum is by definition 1 at all times. Removing one of these shares from the regression could potentially solve the issue with collinearity, however population

shares that are close to one another are typically highly correlated due to the slow-moving nature of population aggregates, leading to unstable parameters. In addition, all results would have to be interpreted relatively to the omitted group rather than on their own, which adds extra complications, and using 17 variables in a regression significantly lowers the number of degrees of freedom.

In order to avoid such issues, I construct a different set of variables employing an alternative method for representing the age distribution, i.e. using a way of parametrising the shape of the population pyramid by fitting a polynomial curve to the population shares by age group. A similar method was first introduced in Fair and Dominguez (1991) (10), and as augmented in Kopecky (2021) (22) it solves both collinearity and correlation issues mentioned above.⁵ In an efficient way of approximating the whole age distribution with a cubic function, the 18 age group population shares can be converted into three demographic variables, D_1 , D_2 and D_3 , that retain information on the shape of the population pyramid. The values of D_1 , D_2 and D_3 for country *i* and year *t* are computed as follows from the 18 age-group population shares:

$$D_{1,i,t} = \left(\sum_{j=1}^{18} j \frac{N_{j,i,t}}{N_{i,t}}\right) - \frac{\sum_{j=1}^{18} j}{18}$$

$$D_{2,i,t} = \left(\sum_{j=1}^{18} j^2 \frac{N_{j,i,t}}{N_{i,t}}\right) - \frac{\sum_{j=1}^{18} j^2}{18}$$

$$D_{3,i,t} = \left(\sum_{j=1}^{18} j^3 \frac{N_{j,i,t}}{N_{i,t}}\right) - \frac{\sum_{j=1}^{18} j^3}{18}$$
(2)

where $N_{j,i,t}$ is the population belonging to age group j in country i and year t and $N_{i,t}$ is total population. These three variables can be used in regression analysis, and the implied coefficients of each age-group population share $N_{j,i,t}/N_{i,t}$ on the dependent variable can be subsequently recovered from the coefficients on D_1 , D_2 and D_3 as further detailed in Section 4.2. Figure 3 shows an example of how the polynomial representation of the demographic distribution based on D_1 , D_2 and D_3 fits the actual population pyramid. Because the fitted polynomial is of third order, at times it fails to capture double humps that would be better approximated by a higher-order polynomial (like in the example), however the fit is very close to the original distribution and the correlation between the original population shares and the implied shares based on the three demographic variables is very high.⁶

⁵See Kopecky (2021) (22) for a detailed explanation of the rationale and derivation of these variables.

⁶The average correlation between the population shares in the original population pyramids and those implied by the D_1 , D_2 and D_3 variables is 94.04%, ranging from 82.80% for the 40 to 44 agegroup to 99.23% for the 5 to 9 age-group. Using a fourth-level polynomial (i.e. adding a D_4 variable) does not qualitatively change any of the results shown below in any meaningful way.

Figure 3: Example of a demographic distribution and its polynomial representation based on the D_1 , D_2 and D_3 variables. USA, 2021 (as from UN World population prospects 2019).



4 Demographics and Rates of Return

4.1 Working-age Population and Productive Capacity

Economic theory and abundant literature suggest that, as the share of population in working age falls, returns on productive capital should follow the same path, as the productive capacity of the economy, per unit of capital, decreases as a result of the decline in labour input. This fact can be shown empirically with a simple exercise. Start from the simple assumptions that the rate of return on capital r^k is a negative function of the capital stock per hour worked K/H and a positive function of productivity A and the capital share of income α :⁷

⁷Intuitively, working a unit of capital for longer (= lower K/H) is going to generate more income, and the same holds if productivity is higher or if a larger share of income is distributed to capital. These conditions hold for instance if we assume a standard Cobb-Douglas production function (with the labour input being hours worked, H) of the form $Y = AK^{\alpha}H^{1-\alpha}$, where we would have $MPK = \alpha \frac{Y}{K} = \alpha A \left(\frac{K}{H}\right)^{\alpha-1}$. However, the above formulation allows for a wider range of potential production functions.

$$r^{k} = f\left(\frac{K}{H}, A, \alpha, ...\right),$$

$$\frac{\partial r^{k}}{\partial (K/H)} < 0,$$

$$\frac{\partial r^{k}}{\partial A}, \frac{\partial r^{k}}{\partial \alpha} > 0$$
(3)

To show the effect of demographic factors on r^k , K/H can be broken down into: $K/N \times L/H \times N/L$, where N is total population, and L is the number of persons employed. In this expression therefore, each factor represents respectively (i) the capital stock per capita; (ii) the inverse of average annual hours worked per person employed; and (iii) the inverse of persons employed as a share of the population. Substituting this into the above equation results in:

$$r^{k} = f\left(\frac{\stackrel{(-)}{K}}{N}, \frac{\stackrel{(-)}{L}}{H}, \frac{\stackrel{(+)}{N}}{L}, \frac{\stackrel{(+)}{\alpha}}{\alpha}, \ldots\right) = f\left(\frac{\stackrel{(-)}{K}}{N}, \frac{\stackrel{(+)}{H}}{L}, \frac{\stackrel{(+)}{N}}{N}, \frac{\stackrel{(+)}{\alpha}}{\alpha}, \ldots\right)$$
(4)

This shows, unsurprisingly, how we should expect a higher share of population at work (L/N) to increase return on capital, and a higher number of hours worked per person employed to also lead to higher returns. L/N can be further decomposed into $L/N_w \times N_w/N$, where N_w is the working-age population. In other words, this corresponds to: (i) persons employed as a share of the working-age population (or the employment rate) times (ii) the share of the working-age population in total population. This finally gives us an expression that directly expresses the return on capital in terms of the working-age population share, where there is a positive relationship between r^k and N_w/N :

$$r^{k} = f\left(\frac{K}{N}, \frac{H}{L}, \frac{L}{N_{w}}, \frac{N_{w}}{N}, \frac{(+)}{A}, \frac{(+)}{\alpha}, \dots\right)$$
(5)

This very simply highlights how, *ceteris paribus*, a higher proportion of the population in their working age should increase the rate of return on capital (all else equal) as it will feed into higher labour supply (hence lower K/H), increasing income.

I want now to test these facts empirically, as all the relevant variables in 4 and 5 can be easily obtained from the data. The above derivation suggests that a regression model of the following form might be appropriate within a fixed-effect panel framework to test the validity of equation 4:

$$\ln(r_{it}^{k}) = \beta_{0} + \beta_{1} \ln(\alpha_{it}) + \beta_{2} \ln(A_{it}) + \beta_{3} \ln(K_{it}/N_{it}) + \beta_{4} \ln(H_{it}/L_{it}) + \beta_{5} \ln(L_{it}/N_{it}) + \gamma_{i} + \delta_{t} + u_{it}$$
(6)

where *i* and *t* represent country and year, γ_i are country fixed-effects and δ_t are year fixed-effects, and all variables are expressed in logarithms. A corresponding equation can be derived for 5. I test 6 using the panel dataset introduced in Section 3 and display the results in Table 1. The estimation employs robust standard errors clustered at country level, and year fixed-effects in specifications (2) and (4). The regression results show that what expressions 4 and 5 suggest in terms of the signs of the effects of each component of K/H, as well as A and α , on r^k holds empirically. As displayed in Table 1, all the assumptions arising from the simple exercise above are confirmed with significance, whether or not year fixed-effects are added, with the coefficients showing the expected signs and being considerably large. Of particular interest for this paper, a greater proportion of the population in their working age appears to have a large and positive effect on return on capital, with a 1% increase in the working-age population to population ratio increasing the return on capital by 0.49% (0.57%).

Although the results in Table 1 are encouraging as they match the equation formulation of the problem, there might be potential issues with such specification. The variables used for the regression in Table 1 are slow-moving, some of them are trending upwards (e.g. K/N, TFP) or downwards (H/L) in most countries, and they might be cointegrated. A visual inspection of the data and the use of unit root tests for the above variables at the country level seem to confirm that for most of the countries, the variables included in the above specifications are generally non-stationary, and in most of cases they have a trend. (Panel unit root tests are more contradictory in this regard however, with some types of tests rejecting the null hypothesis that the variables are non-stationary in all panels for some of the variables.) There is a strong possibility that the variables might therefore be cointegrated, and indeed in the case of a few countries, a Vector Error-Correction model with one cointegrating equation would well fit the data.

To address these issues I have separately estimated specifications (1) and (3) of Table 1 (i) with all variables differenced once, and (ii) as an Arellano-Bond dynamic panel. These give similar results: in both regression models, the coefficients on all the explanatory variables remain highly significant, large and with the expected signs. The results from Table 1 do not appear to be the consequence of a spurious relationship, and can be reasonably taken as valid.⁸

Another concern lies in the choice of variable for productivity. To generate a measure of Total Factor Productivity, as calculated in PWT and used for Table 1, a certain underlying

⁸Results are available but are omitted for brevity.

	(1)	(2)	(3)	(4)
	$ln(r^k)$	$ln(r^k)$	$ln(r^k)$	$ln(r^k)$
	eta (s.e.)	eta (s.e.)	eta (s.e.)	eta (s.e.)
$ln(\alpha)$	0.909***	0.930***	0.909***	0.932***
	(0.09)	(0.09)	(0.09)	(0.09)
ln(TFP)	0.673***	0.614***	0.674***	0.617***
	(0.09)	(0.10)	(0.09)	(0.10)
ln(K/N)	-0.352***	-0.322***	-0.354***	-0.333***
	(0.04)	(0.05)	(0.04)	(0.05)
ln(H/L)	1.061***	0.902***	1.058***	0.879***
	(0.23)	(0.26)	(0.23)	(0.26)
ln(L/N)	0.478***	0.479***		
	(0.11)	(0.13)		
$ln(L/N_w)$			0.471***	0.442***
			(0.14)	(0.14)
$ln(N_w/N)$			0.492***	0.570***
			(0.16)	(0.19)
Constant	-4.804**	-3.878*	-4.753**	-3.540
	(1.98)	(2.09)	(1.99)	(2.13)
Country fixed-effects	\checkmark	\checkmark	\checkmark	\checkmark
Year fixed-effects		\checkmark		\checkmark
R^2	0.632	0.656	0.632	0.657
deg. fr.	63	63	63	63
# obs.	3000	3000	3000	3000
$ln(H/L)$ $ln(L/N)$ $ln(L/N_w)$ $ln(N_w/N)$ Constant Country fixed-effects $\frac{R^2}{\text{deg. fr.}}$ # obs.	(0.04) 1.061*** (0.23) 0.478*** (0.11) -4.804** (1.98) √ 0.632 63 3000	(0.05) 0.902*** (0.26) 0.479*** (0.13) -3.878* (2.09) √ √ 0.656 63 3000	(0.04) 1.058*** (0.23) 0.471*** (0.14) 0.492*** (0.16) -4.753** (1.99) √ 0.632 63 3000	(0.05) 0.879*** (0.26) 0.442*** (0.14) 0.570*** (0.19) -3.540 (2.13) √ √ 0.657 63 3000

Table 1: Fixed-effects panel estimation of return on capital based on equations 4 and 5.

production function in the economy needs to be assumed; this production function might not be the true underlying function and bias the results through the addition of TFP in the regression. At the same time, omitting TFP from the regression would clearly lead to omitted variable bias. In the absence of any specific reason for choosing a different productivity measure (which would still be reliant on some different assumption), and because of its widespread and established use in the literature, in the analysis that follows I will continue using TFP as produced by the Penn World Tables. However, in Appendix A.1, I calculate alternative productivity measures and repeat the exercise done for Table 1 (as well as the following Section). The results suggest that using TFP remains appropriate.

4.2 Age Distribution and Capital Accumulation

While the above analysis suggests that the working-age population share alone has a direct impact on returns due to productive capacity considerations, economic theory also suggests that the *structure* of the population, i.e. the shape of the population pyramid, is in turn likely to affect some of the other factors into which the return on capital was decomposed in equations 4 and 5. Ignoring these channels can potentially significantly over- or underestimate the effect of the demographic distribution on returns. Indeed, even among people of working age, agents of different ages can have quite different effects on the economy.

Every item in equation 4 might be dependent on demographics. In particular, K/N stands out in this sense. An abundant literature on the consumption, savings, and wealth accumulation patterns of households throughout their life cycle shows how households typically start their working life with limited assets (or in debt); they gradually build up their wealth through the years by accumulating savings from their incomes, which also typically rise through the working years, reaching a peak after their fifties; finally, once they retire, they slowly run down their assets. Such a pattern of behaviour has been frequently observed empirically.

While every person's wealth accumulation journey is different and mostly independent, this pattern of behaviour aggregates across the population such that if an economy's population, all else equal, has relatively more people of an age at which life-cycle wealth is at its peak, then the overall level of capital per person in the economy will be higher. Note how, as K/N came with a negative coefficient on returns in Table 1, that implies that the return on capital will be lower in such an economy due to the abundance of capital relative to the population, which is in turn due to the simultaneous large holdings of wealth by a large cohort of individuals.

In addition to capital per person, the demographic distribution is likely to have a major impact on the employment rate of the population (L/N) as well. Different age groups have vastly different labour force participation rates, with some groups not participating in the

labour force at all, while even within the working-age population, individuals of different ages have different likelihoods of being in employment.

Finally, although Total Factor Productivity, or its growth, might also be higher *ceteris paribus* with a relatively younger workforce, as has been argued before in the literature (see Section 2), such a relationship with the age distribution is less clear and established, and therefore TFP is taken as exogenous to the demographic distribution in the analysis that follows.

To obtain an estimate of the combined effect of the age distribution on the rate of return, while accounting for the wealth accumulation and employment rate channels (i.e. the effects of the age distribution on K/N and L/N), I estimate a panel instrumental variable regression based on specification (2) of Table 1, where however ln(K/N) and ln(L/N) are instrumented by the demographic variables D_1 , D_2 and D_3 (calculated as explained in section 3.3), plus year fixed-effects.⁹ Table 2 displays the results of the first-stage regressions on ln(K/N) and ln(L/N).

	(1)	(2)
	ln(K/N)	ln(L/N)
	b/se	b/se
D_1	0.260	0.731***
	(0.21)	(0.09)
D_2	0.097***	-0.046***
	(0.03)	(0.01)
D_3	-0.006***	0.000
	(0.00)	(0.00)
ln(TFP)	0.800***	-0.118***
	(0.02)	(0.01)
ln(lpha)	-0.099***	0.036***
	(0.03)	(0.01)
ln(H/L)	0.924***	-0.104***
	(0.08)	(0.03)
Constant	3.851***	-0.309***
	(0.58)	(0.24)
Country fixed-effects	\checkmark	\checkmark
Year fixed-effects	\checkmark	\checkmark
R^2	0.919	0.648
# obs.	3000	3000

Table 2: Instrumental-variables estimation of r^k . First-stage regressions.

The demographic variables D_1 , D_2 and D_3 are jointly significant at the 1% level for both

⁹The shape of the population pyramid is an ideal instrument, as an economy's population structure is almost entirely independent of *current* economic conditions, due to the very slow-moving nature of population dynamics. Even migration and major events that affect both the economy and demographics (pandemics such as COVID-19, for instance), have a negligible impact on the population pyramid.

ln(K/N) and ln(L/N), indicating that the shape of the population pyramid that they represent has itself an impact on these variables. The implied coefficients of each age bracket on ln(K/N) and ln(L/N) can be recovered from the coefficients on D_1 , D_2 and D_3 , based on how these were calculated from the relative sizes of the age brackets (see Section 3.3). Recalling that there are 18 age brackets in the data (17 5-year age groups, plus a group including all the population above 85 years old), the implied coefficient $\beta^{j,ln(K/N)}$ of age group j on ln(K/N), given the coefficients $\beta^{D_1,ln(K/N)}$, $\beta^{D_2,ln(K/N)}$ and $\beta^{D_3,ln(K/N)}$ from the first-stage of the instrumental variables regression in Table 2, can be calculated as:

$$\beta^{j,ln(K/N)} = -\left(\frac{\beta^{D_1,ln(K/N)}}{18}\sum_{k=1}^{18}k + \frac{\beta^{D_2,ln(K/N)}}{18}\sum_{k=1}^{18}k^2 + \frac{\beta^{D_3,ln(K/N)}}{18}\sum_{k=1}^{18}k^3\right) + j\beta^{D_1,ln(K/N)} + j^2\beta^{D_2,ln(K/N)} + j^3\beta^{D_3,ln(K/N)}$$
(7)

Figure 4: Effect of each age group on ln(K/N) and ln(L/N)



with j = 1 representing the age group of persons aged 0-4 years old, j = 2 representing 5 to 9 year-olds, and so on at 5-year increments. The equivalent can be done for ln(L/N). Figure 4 displays graphically the backed-out coefficients from the first-stage regressions for each age group population share, derived from the coefficients on D_1 , D_2 and D_3 as for the equation above. The figure shows how, in accordance with the life-cycle hypothesis, agents in their late working-age and early retirement increase capital per person in the economy, as these are the years in which the highest level of wealth is accumulated before asset positions begin to be sold as individuals age further; conversely, a larger proportion

of children and the elderly in the population tends to decrease capital per capita. Unsurprisingly, the coefficients on the employment rate are instead positive for all age groups of working age. The peak positive impact of an age-group on employment comes earlier than for capital accumulation: while 55 to 64 year-olds have the largest impact on ln(K/N), 35 to 44 year-olds have the largest impact on employment.

	(1)
	$ln(r^k)$
	b/se
ln(K/N)	-0.443***
	(0.02)
ln(L/N)	0.854***
	(0.06)
ln(TFP)	0.773***
	(0.04)
ln(lpha)	0.901***
	(0.02)
ln(H/L)	1.015***
	(0.06)
Constant	-3.148*
	(0.44)
Country fixed-effects	\checkmark
Year fixed-effects	\checkmark
# obs.	3000

Table 3: Instrumental-variables estimation of r^k . Second-stage regression.

The second-stage of the IV panel regression (Table 3), with ln(K/N) and ln(L/N) instrumented by D_1 , D_2 , D_3 and year fixed-effects, gives results that are very similar to those from specifications (1) and (2) in Table 1. However, this specification allows to produce an estimate of the effects of the whole age distribution on $ln(r^k)$. By multiplying the implied coefficients of the age groups on ln(K/N) and ln(L/N) from the first-stage regressions as displayed in Figure 4, by the coefficients of ln(K/N) and ln(L/N) themselves on the rate of return from the second-stage regressions, we can estimate the effect of each age group on the return on capital through wealth accumulation channel, and the employment channel. The sum of the two (see equation 8) provides an estimate of the overall impact of each age group on rates of return through the combined effect of these two channels. The results are displayed graphically in Figure 5.

$$\beta^{j,ln(r^k)} = \beta^{j,ln(K/N)} \beta^{ln(K/N),ln(r^k)} + \beta^{j,ln(L/N)} \beta^{ln(L/N),ln(r^k)}$$
(8)

These results are very intuitive. Not participating in the labour force, the effect of children and retired persons on returns via the labour force participation channel is negative. At the



Figure 5: Combined effect of each age group on rates of return via ln(K/N) and ln(L/N)

same time, children and young adults, and persons in late retirement contribute a positive effect on returns via capital accumulation due to their low levels of accumulated wealth, for opposite reasons. This results in an overall positive effect on returns by young agents overall, with the peak positive impact at about 20 years of age and then steadily decreasing.

As agents enter the labour force, they increase returns to capital through their labour, but while their savings gradually accumulate, steadily increasing K/N, the negative impact on returns from wealth accumulation rises and finally overcomes the positive effects of labour force participation in the 45-49 age bracket, after which the combined effect becomes negative. With the labour-force participation effect now falling steadily and wealth accumulation remaining very high, the 70-79 age groups become the ones that most adversely affect returns. While agents above 80's typical asset decumulation leads to a positive effect on returns through the saving channel, this is not sufficient to fully compensate for the negative effect through the employment channel.

The median ages of advanced economies are now moving towards the peak earning years of the late working age (> 50), while the shares of people in their teens, twenties and thirties are rapidly falling due to falling birth rates, and the share of over-65s rises steadily. This results in lower returns on capital due to the combined effects of (i) a falling working-age population overall, and especially the younger end; and (ii) a higher proportion of people with substantial wealth, relatively less productive and close to retirement bringing up K/N.

Based on this analysis and projected demographic developments across the world, advanced economies are set to continue to experience lower and falling rates of return for many

years, while an increasing number of developing economies which are starting to experience a demographic transition will also see their rates of return fall in the next decades.

These factors highlight a stark generational imbalance between those currently in or close to retirement, who were able to build up considerable amounts of wealth in the past, when the younger structure of the population, high growth and relative scarcity of capital allowed for high returns; and a cohort of young individuals facing a future of low returns impairing their ability to build wealth themselves. The effects of falling returns on intergenerational wealth imbalances are explored further in the next Section.

4.3 Returns to Wealth and Inequality Between Generations

The previous Section showed how a relatively young population positively affects rates of return on capital, as in young economies labour is abundant and capital relatively scarce. Similar results are well-established in the literature on secular stagnation, in particular with respect to the effects of population ageing on long-term equilibrium interest rates and risk-free rates.

In an ageing economy, as rates of return gradually decline, including due to demographic factors, and wealth becomes increasingly concentrated in the hands of older age groups, the accumulation of wealth becomes more and more difficult for younger generations of workers. At the same time, the lower production of income out of wealth due to declining rates puts an increasing number of retirees with limited resources into financial difficulties in the absence of generous, but increasingly expensive public pension schemes. In such an economy, production by the current, shrinking young labour force needs to generate enough returns and tax revenues to sustain the incomes of a rising number of retirees, through public pension schemes as well as capital gains.

To illustrate the link between population dynamics and wealth inequalities between generations, the results from Section 4.2 can be used to compute the overall effect of the aggregate demographic distribution of an economy on r^k , and therefore on total rates of return in the long term. To do so, for every country and year, the implied coefficients of each age bracket on $ln(r^k)$, as illustrated graphically in Figure 5, are multiplied by the share of each age bracket in the total population, to give the impact of such age bracket on $ln(r^k)$. These are then summed up in order to recover the impact of the aggregate demographic distribution itself. The effect of the demographic distribution on $ln(r^k)$ in year t on country i is therefore calculated as:

$$\beta_{i,t}^{DD} = \sum_{j=1}^{18} \beta^{j,ln(r^k)} \frac{N_{j,i,t}}{N_{i,t}}$$
(9)

where $N_{j,i,t}$ is population in age bracket j in country i and year t ($N_{i,t}$ is the total population in the same country and year), and the β^{j} 's are calculated from the coefficients on D_1 , D_2 and D_3 as in equation 7.

By using the values for $\beta_{i,t}^{DD}$ across countries and years, I can estimate the differential impact of age distributions on the accumulation of wealth of different generations, and explore the extent to which the impact of age distributions on returns may represent a significant source of wealth inequality between generations, in the present and future.

Figure 6: Impact of the demographic distribution on return on capital (r^k), selected advanced economies.



Figure 6 displays the estimated effect of the overall demographic distribution on returns on capital for a number of advanced economies. This is computed from $\beta_{i,t}^{DD}$ as per equation 9; as $\beta_{i,t}^{DD}$ is the effect of the demographic distribution on the natural logarithm of r^k , the effect on r^k itself is calculated as:

$$DD_{i\,t}^{effect} = r_{i\,t}^k - e^{ln(r_{i,t}^k) - \beta_{i,t}^{DD}}$$
(10)

While it turns out that such an effect has been considerably positive for most of the time series and countries under consideration, due to young and growing populations throughout most of the twentieth century, the trend is downwards everywhere (including countries not in the figure), with aging populations and lower birth rates moving the demographic distribution towards age groups that lower r^k .

Figure 7 further illustrates the evolution of rates of return in China, Italy, Japan and the





USA,¹⁰ the estimated impact of the demographic distribution on these returns throughout the years, and finally what returns would have been in the absence of such demographic factors. As is evident for the countries displayed in Figure 7¹¹, rates of return after removing demographic effects (i.e., $r^k - DD_{i,t}^{effect}$) appear to be significantly more stable throughout the years in individual economies (with little to no sign of them trending up or downwards in most countries), and also more similar between economies, as countries with higher rates overall also tended to have younger populations.¹² Thus, for a large part due to demographic developments, rates of return saw a significant decline in recent decades as the positive contribution of the population structure (a "demographic dividend" that certain generations have enjoyed) fell across the world. In a few countries (e.g. Italy and Japan), the demographic distribution is now weighing on rates of return, which would have been higher otherwise, with the turning point occurring in both cases soon after the year 2000. As also shown in Figure 6, more economies are now in this position due to their ageing pop-

¹⁰Results for the other countries are available. These countries were chosen because they include two of the oldest populations in the world (Italy and Japan), and the two largest economies (USA and China). The case of the USA is also interesting as it remains one of the youngest advanced economies (partly due to high immigration) and did not experience as dramatic a demographic shift as most of Europe; on the other hand, the population of China is aging at a fast rate also due to its former one-child policy.

¹¹But also for the rest of the countries

 $^{^{12}}$ The standard deviation of r^k across the entire sample is 0.083, while for r^k with demographic effects removed it reduces to 0.052.

ulations, or are trending towards it.

Figure 7 also presents a forecast of the evolution of rates of return through the end of the century. This is done by assuming that $r^k - DD_{i,t}^{effect}$, which as discussed has been relatively constant throughout the sample period, remains at a level equal to its country average from the year 2000 to the latest observation. $DD_{i,t}^{effect}$ is then calculated by obtaining D_1 , D_2 and D_3 based on the UN population projections. r^k is then finally the sum of the two. In countries where the demographic transition is already at an advanced stage, rates of return are expected to fall further (but only slightly) for a few more decades, reaching their lowest point at around 2040/2050, to then stabilise around that level, below the rate of return that would have been achievable in the absence of any demographic contribution, and significantly below the rates of return experienced in the second half of the twentieth century. Economies at an intermediate stage of the demographic transition (e.g. China) are expected to follow the same path with a few decades' delay; their rates of return are seen declining rapidly in the next years.

Most developing economies are still expected to enjoy a high positive contribution to returns, thanks to their relatively young population, for some time to come; however, in almost all cases, rates of return are expected to decline significantly and steadily for the most part of the century. The contribution of the population of China to China's rate of return on capital, for instance, stood around 2.9% in 2017, while it is expected to reach a level of -0.7% by around 2070; this implies a decline in rates of return there of almost 7 basis points per year for over 50 years, due to a rapidly declining birth rate and rising life expectancy. Finally, countries which are expected to maintain relatively higher birth rates and/or higher levels of net migration (e.g. the US) will see rates of return declining at a much slower pace.

Young, growing and dynamic populations throughout the twentieth century have helped maintain high returns to capital/wealth, thanks to a high share of young and productive workers and a relative scarcity of capital; these high returns have allowed generations of middle-class households to accumulate sufficient wealth for a decent retirement. Indeed, throughout this time, a pyramid-shaped population distribution has sustained a pyramid scheme of sorts, where the returns on wealth of a cohort were guaranteed by income produced by the next, larger and younger cohort of workers.

As life expectancies increased, and birth rates fell, population pyramids in advanced economies started stabilising, as new generations were no longer larger than the previous ones and more people survived to older ages; in recent years, population pyramids have started to become inverted, with younger generations shrinking more and more. Recall Figure 1 and the developments in the population of Japan, where the demographic shift is currently at the most advanced stage globally. In Japan and economies with similar demographic developments, what was before a "pyramid scheme" where increasingly large younger generations supported the retirement incomes and wealth of a small older population, has slowly

morphed into an *inverted pyramid scheme*, where it becomes quite hard for a shrinking young population to now do the same.

Whether an agent is young, middle aged, or retired, the age of everyone else around them matters greatly to their wealth accumulation, via the effects of the demographic distribution on returns on capital as seen above. But how much more wealth could an agent who started saving in the 1960s, with a growing and young population, accumulate over the next 20 years compared to a similar agent who started investing in the 2000s, and how much of this difference can be attributed to demographics?

If returns are compounded annually, a small difference in annual returns can lead to very significant wealth accumulation paths across a working life. With the "demographic dividend" described here starting to fall across the developed world around the 1970s and bringing down rates of return with it, each subsequent generation would have been at a disadvantage compared to the previous one in terms of their wealth accumulation. Newer generations would either accumulate less wealth out of the same savings compared to their elders, or be forced to save a larger portion of their incomes to make up for the impact of the lost returns. Figure 8 shows how a small difference in the rates of return experienced by households in subsequent generations in the countries of Figure 7 lead to widely diverging wealth accumulation journeys. An Italian household who invested 1 in 1955 and reinvested the return (r^k) every year for 20 years could expect to have accumulated a real value of 30 by 1975. By 2017, an investment of 1 20 years prior would have only returned a real value of 10. Notice how this wedge was mostly due to the very favourable demographic structure just after WWII pushing up rates of return, which then fell steadily from the 70s as population started ageing. In the absence of these demographic effects, the two households would have enjoyed virtually the same cumulated returns out of their investments.

This paper largely focuses on how demographic dynamics unequally impact wealth accumulation for different generations or cohorts. However, the large imbalances between generations in the speed of wealth accumulation have an impact on wealth inequality through multiple different channels.¹³

The first is apparent from Figure 8. In general, as they more easily accumulated wealth while younger, older generations will hold a larger share of wealth. On the other hand, younger people will be further away from the average per-capita wealth level than their elders were at the same age. This leads to *inter*-generational wealth inequalities.

At the same time, falling rates of return might mitigate *intra*-generational wealth inequalities. With high rates of return, a small difference in savings between two agents of the same age can compound to become very large over decades; with lower returns, this is less of the

¹³One of these is the decline in rates of return documented here. Greenwald et al. (2021) (18), for instance, find that declining real interest rates lead to an increase in financial wealth inequality.

Figure 8: Wealth accumulated over 20 years: CHN, ITA, JPN, USA



The figure shows, for each year, the wealth accumulated if a 1 unit investment was made 20 years before, and returns were each year reinvested and compounded at the prevailing rate of return. The rates of return considered are the r^k and r^k with demographic effects removed, as in Figure 7.

case, leading to less wealth inequality between agents of the same age.¹⁴

Lower rates of return also tend to raise asset prices. Indeed, while Figure 2 showed how the price of capital remained unchanged overall, this was due to a combination of falling prices of machinery, transport equipment and other forms of capital, while the price of buildings and residential structures (and with it the value of housing assets) kept increasing in real terms. With the possession of housing assets very low at young ages and low wealth percentiles, the appreciation of housing assets due to falling rates, which are in turn due to demographic developments, would increase both intra- and inter-generational wealth inequalities.

While the above is true if we consider wealth as a single aggregate asset, agents of different ages tend to invest in different asset types, such that a different demographic distribution is likely to affect the relative prices and returns of different asset types. Kopecky and Taylor (2020) (23) suggest that aging demographics raise the equity risk premium; at the same time, housing has become more expensive in the past decades as seen in Section 3.2. As

¹⁴Note, however, that regardless of initial wealth, agents would be worse off, with lower wealth (all else equal) than in the presence of higher returns. Also, this is only the case if agents invest in the same portfolio. With different portfolio compositions, agents investing in higher-return assets might benefit if risk premia increase with lower rates of return.

agents who invest in these asset types tend to be wealthier, a falling rate of return on capital (aggregate) may affect agents in the lower end of the wealth distribution more than those in the upper end, further increasing wealth inequalities.

Intergenerational transfers (bequests) further complicate the picture. An increasing proportion of wealth being held by the elderly signifies that increasing amounts of wealth will be ultimately transferred to younger generations. In a young economy, transfers might decrease inequality by redistributing wealth to a higher number of less wealthy agents. A longer life expectancy, however, means that the average age at which agents receive such a transfer increases further.¹⁵ With life expectancy at over 80 in many advanced economies, the age at which such transfers occur would tend to fall during an agent's late working age, i.e. at the peak of their wealth accumulation, further concentrating wealth in those age groups.¹⁶ Lower birth rates also mean that wealth transfers will be increasingly concentrated to a decreasing average number of heirs. All in all, in an ageing society bequests are thus likely to worsen inter-generational wealth inequality (as transfers tend to go to already wealthy age groups), as well as intra-generational wealth inequality (transfers are more concentrated).

The analysis presented so far points to convincing evidence of population dynamics having a significant effect on returns, and consequently on the unequal distribution and accumulation of wealth, especially between different generations in ageing societies. The empirical evidence from this section suggests that decreasing rates of return as population ages lead to rising inter-generational inequality, if subsequent generations invest similar amounts to each other. However, does the situation change when wage levels are taken into account? The next section explores this.

5 Demographics and Wages

We have seen that population ageing contributes to a decrease in rates of return. This leads to divergences in how much wealth a household can expect to accumulate out of a certain level of savings. At parity of (real) investment, current young generations are disadvantaged compared to previous generations in terms of wealth accumulation, due to lower rates of return. However, the situation may change if wages are taken into consideration:

¹⁵This effect is slightly mitigated by the fact that parents' age at a child's birth has also tended to be on the increase in ageing economies. However, life expectancies have risen by longer, meaning that, overall, the average age of receipt of bequests has risen: for instance, Eurostat data (available at https://ec.europa.eu/eurostat/web/main/data/database) shows that the mean age of women at childbirth in Belgium has risen from 28.0 years in 1960 to 30.8 in 2020; over the same period, life expectancy increased from 69.7 to 80.8 years. Similar trends can be seen in other countries.

¹⁶Intergenerational gifts when the recipients are in their working age can alleviate these concerns, although these tend to be smaller than bequests.

since the 1950s, real wages have steadily increased across the world, such that in most economies an average young household today can afford to invest more (in real terms) than previous generations. Net of this, are younger generations better or worse off?

A large part of increases in labour compensation in past decades can be attributed to technological advancement and higher productivity. However, the same demographic dynamics that have been discussed for returns on capital also have a bearing on the compensation of labour, where for the most part their impact on wages is opposite to their impact on returns. In particular, a smaller proportion of the population at work and a higher capital intensity both tend to lead to higher wages.

The exercise in Section 4.1 can be repeated for wages (per hour worked). With similar arguments as for capital, I make three assumptions about the return on labour which are consistent with 3 and economic theory, i.e. wages will be higher if: (i) there is more capital relative to labour K/H (with more capital, an hour of labour can produce more); (ii) productivity is higher; (iii) the capital share of income is lower. In other words:

$$w^{H} = f\left(\frac{K}{H}, A, \alpha, ...\right),$$

$$\frac{\partial w^{H}}{\partial (K/H)}, \frac{\partial w^{H}}{\partial A} > 0,$$

$$\frac{\partial w^{H}}{\partial \alpha} < 0$$

(11)

Decomposing K/H as in 4 gives:

$$w^{H} = f\left(\frac{\overset{(+)}{K}, \overset{(-)}{L}, \overset{(-)}{L}}{N}, \frac{L}{N} \left(= \frac{\overset{(-)}{L}, \overset{(-)}{Nw}}{N} \right), \overset{(+)}{A}, \overset{(-)}{\alpha}, \ldots \right)$$
(12)

i.e., contrary to the case for rates of return on capital, we expect K/N to increase hourly wages and a higher proportion of the population at work to decrease them. The average total wage earned by a worker, w^L , would be equal to $H/L \times w^H$. Assuming that the elasticity of hourly wages to hours worked per person is less than one (workers would not be working as many hours if that was not the case), we can expect H/L to have a positive effect on w^L such that:

$$w^{L} = f\left(\frac{\overset{(+)}{K}}{N}, \frac{\overset{(+)}{L}}{L}, \frac{\overset{(-)}{L}}{N}\left(=\frac{\overset{(-)}{L}}{N_{w}}\frac{\overset{(-)}{N}}{N}\right), \overset{(+)}{A}, \overset{(-)}{\alpha}, \ldots\right)$$
(13)

How do demographics affect wages through the channels in equation 13? We know that ageing demographics tend to increase K/N and lower L/N (recall Figure 4). Running the

same instrumental variables regression as in section 4.2 (replacing r^k with w^L as dependent variable) unsurprisingly gives results that are consistent with equation 13, as shown in Table 4. The first-stage regression results are still the same as for Table 2 (the variables entering the first-stage regressions are the same), giving the age-profile of effects on K/N and L/N as per Figure 4.

	(1)
	$ln(w^L)$
	b/se
ln(K/N)	0.566***
	(0.01)
ln(L/N)	-0.229***
	(0.04)
ln(TFP)	0.792***
	(0.02)
$ln(\alpha)$	-0.800***
	(0.01)
ln(H/L)	0.702***
	(0.03)
Constant	-2.552***
	(0.22)
Country fixed-effects	\checkmark
Year fixed-effects	\checkmark
# obs.	3000

Table 4: Instrumental-variables estimation of real (log) w^L . Second-stage regression.

As done above for Figure 5, multiplying these implied coefficients of each age group on ln(K/N) and ln(L/N) by the coefficients of ln(K/N) and ln(L/N) from the second-stage regression, I can calculate the impact of each age group on labour compensation ($ln(w^L)$) through wealth accumulation, through the labour force, and the overall effect. Figure 9 shows this graphically. The strongest impact is through wealth accumulation, where age groups with high accumulated wealth lead to an abundance of capital in the economy, which generates an increase in the compensation of labour by making it more productive.

A higher employment rate of the population has the opposite effect, as less scarcity of labour pushes down its cost; this channel seems however to be smaller than the wealth accumulation channel through demographics.¹⁷ Overall, all age groups from 35-39 to 80-84

¹⁷Note how in Figure 4, the effects of the age distribution on ln(L/N) are flatter than those on ln(K/N). This is partly because, in ageing economies, the labour force participation of the working-age population has been observed to increase (at least in part due to demographic developments) as the working-age population share itself shrinks; at the same time, labour force participation in groups outside of the typical working-age population (over 65s) has also tended to rise in these economies, also because of improving health. The overall effect of demographics on the employ-

Figure 9: Combined effect of each age group on compensation of labour via ln(K/N) and ln(L/N)



lead to higher wages, all other things constant, while over-85s and under-35s reduce them. As a result, ageing economies tend to have higher wages due to demographic factors, and in particular due to the effect of the wealth accumulated by those from middle age to early retirement. Comparing this with Figure 5, which shows the effect of each age group on return on capital, we can see that for most age groups, having a positive impact on return on capital means having a negative impact on compensation of labour, and vice-versa. However, crucially, the magnitudes are not quite the same for all age groups (for instance, while the 65-69 age group has a large negative impact on return on capital and a large positive impact on compensation, the 45-49 group has a high positive impact on wages but only a negligible negative impact on returns); additionally, persons aged 35-44 have a positive effect on both returns and wages, while the opposite is true for over-85s.

The overall effect of the population pyramid on the compensation of labour is shown in Figure 10. More specifically, the overall coefficient of the demographic distribution on $ln(w^L)$, $\beta_{i,t}^{DD}$, is calculated as in equation 9. From this (due to the fact that the regression is in log form), the effect of demographics on actual real wages (in constant 2011 US dollars) can be calculated as:

$$DD_{i,t}^{effect} = w_{i,t}^{L} - e^{\ln(w_{i,t}^{L}) - \beta_{i,t}^{DD}}$$
(14)

ment rate, while certainly strong, is therefore not as stark as might seem.

where $w_{i,t}^L - DD_{i,t}^{effect} = e^{ln(w_{i,t}^L) - \beta_{i,t}^{DD}} = w_{i,t}^{Lnodem}$ can be seen as the level of labour compensation that would have prevailed in the absence of any effect due to demographics. Figure 10 shows how big $DD_{i,t}^{effect}$ is as a proportion of this $w_{i,t}^{Lnodem}$. We can see that in the 1950s to 1980s, young population pyramids, leading to higher proportions of the population at work and lower accumulated wealth per capita,¹⁸ caused wages to be up to 30 (and more) per cent lower than would otherwise have been the case. As populations in advanced economies aged, this negative effect has progressively been reducing and has now turned positive in most of these economies; in some of them, wages would now be up to 20% lower if it were not for the scarcity of labour and abundance of capital that is due to demographic dynamics.





According to this it appears that, in the most recent years, the impact of demographics has been the main factor behind increases in real wages in advanced economies. Productivity growth has slowed down considerably, and any potential increases in annual wages per worker due to rising productivity have been in many cases almost exactly compensated by a decrease in average hours worked. On average for these countries, real wages, once demographic effects are removed, have remained unchanged since around 2000. Figure 11 displays this.

¹⁸Note that this is after accounting for long-run trends in capital accumulation, labour force participation and productivity, which are dealt with in the regressions.

Figure 11: Evolution of (log) real wages per person employed (actual versus with demographic effects removed), selected advanced economies.



Note: Values are in log constant 2011 US dollars. The countries included are the same as for Figure 10.

6 Wealth Accumulation: Winners and Losers

The previous sections have detailed how demographic dynamics have contributed to a decline in rates of return while positively affecting wages. Also due to this, the experiences of subsequent generations in ageing economies have been (and will be) substantially different. This section gathers the results shown above for returns and wages and seeks to quantify the overall effect of demographic change on the wealth accumulation journeys of different generations in economies across the world, with a look at future decades. Are there generations that stand to lose or to gain due to these dynamics? What are the implications in terms of inter-generational wealth inequality?

In an environment of increasing wages but decreasing rates of return, subsequent generations are able to save more, but can also expect to *have* to save more in order to accumulate a certain level of wealth, because returns are lower. To determine what effect dominates in any given year and country, I perform a couple of simple exercises that intuitively illustrate how conditions for wealth accumulation have changed through time and across country, and the role of demographics. Similar to the exercise done for Figure 8, I first look at the wealth accumulated out of an investment made at a point in time, but also taking the wage level into account. Specifically, I calculate how much real wealth ($W_{i,t}^{acc}$) is accumulated by time t+20 out of an investment of a fraction s_r of the average wage at time t, if this investment returns the relevant prevailing rate of return (r^k) every year, and all yields are reinvested.

$$W_{i,t}^{acc} = s_r w_{i,t}^L \prod_{j=0}^{19} (1 + r_{i,t+j}^k)$$
(15)

Compare this to a hypothetical situation where there is no effect of demographics, i.e. the wage level is $w^{L,nd}$ and the return on capital is $r^{k,nd}$. The wealth accumulated would then be:

$$W_{i,t}^{acc,nd} = s_r w_{i,t}^{L,nd} \prod_{j=0}^{19} (1 + r_{i,t+j}^{k,nd})$$
(16)

After the end of the PWT sample, estimated demographic effects can be computed thanks to the UN population projections spanning until 2100. However, a forecast for $w^{L,nd}$ and $r^{k,nd}$ needs to be made.¹⁹ As above, I take $r^{k,nd}$ as constant through 2100 for each country given its stability in the sample. In terms of wages, I formulate three very different stylised scenarios (with very different implications), with a focus on the prospects of ageing advanced economies:²⁰

- (1) $w^{L,nd}$ remains constant. This scenario is possibly the most plausible for advanced economies as it has arguably been the case since 2000 (compare Figure 11). The average change in $w^{L,nd}$ since 2000 in advanced economies is 0%.²¹
- (2) It grows at a level equal to its average growth (across advanced economies, unweighted) since 1980. This is equal to around one third of a percentage point per year. The average trend Total Factor Productivity growth across the countries (including developing) in the sample is less than half this, while the average TFP growth since 2000 is around 0.5%. This scenario can represent an average country, although this growth rate is slightly optimistic for advanced economies given the slowdown in productivity in recent decades.
- (3) In each country, it grows linearly up to 2100 to reach the highest level (across

¹⁹ w^L and r^k will then be the forecasted $w^{L,nd}$ and $r^{k,nd}$ plus the respective demographic effects calculated from D_1 , D_2 and D_3 .

²⁰No assumption about developments so far in the future can be justified, so the approach is to try and cover as many options as possible, rather than forecast.

²¹If the vast literature on secular stagnation is to be heeded, growth is unlikely to pick up significantly in the next decades, and there are several culprits, including demographic change.

countries) reached at the end of the PWT sample.²² This implies no or little growth for the countries closest to that level in 2017, and a very quick catch-up for those further away (most of them). This scenario would be representative if we assume convergence, although possibly a stretch in many cases, including for several advanced economies.²³

Figure 12: Wealth accumulated over 20 years from initial investment of 10% of wage, selected advanced economies.



Note: Expressed as a fraction of its peak value up to 2017. The countries included are the same as for Figure 10.

Figure 12 plots the evolution of $W_{i,t}^{acc}$ for a selection of advanced economies under the three scenarios for $w^{H,nd}$, assuming a saving rate of 10%. Because of large differences in levels across countries and since we are interested in the relative increase or decrease in $W_{i,t}^{acc}$ over the years, this measure is normalised by dividing by the country's maximum observed $W_{i,t}^{acc}$ before the end of the sample, i.e. a value of 1 corresponds to the country's peak level of $W_{i,t}^{acc}$ up to 2017. To better see the trends, Figure 13 displays the simple average of each set of lines in Figure 12. A few facts appear evident from these figures:

 $^{^{22}}$ The country with the highest $w^{L,nd}$ in 2017 was Norway.

²³For instance, the country in the 75th percentile of the distribution in the sample in 2017 had an $w^{L,nd}$ which was about 45% lower than Norway's, implying a growth rate well above (2) over the forecast horizon. The median country had an $w^{H,nd}$ 63% lower which implies an even higher growth rate.

Figure 13: Wealth accumulated over 20 years from initial investment of 10% of wage, as a fraction of peak value, average across selected countries.



Note: Averages of each set of lines in Figure 12.

- Thanks to robust technological and wage growth, wealth accumulation since 1950 has on average become steadily faster until around the year 1990, while it has declined significantly since 2000.
- Under the two more plausible scenarios for advanced economies, wealth accumulation in future decades is set to become harder or remain at current levels, which are lower than a generation ago but significantly higher than before the 1980s.
- Where $w^{H,nd}$ and $r^{k,nd}$ are both unchanged (scenario (1)), the impact of demographic dynamics on wealth accumulation through returns prevails on the impact through wages. In general, for most countries across all three scenarios, in the absence of demographic effects $W_{i,t}^{acc}$ would either not decrease as steeply or it would increase more strongly (i.e., demographic factors weigh down on the growth of $W_{i,t}^{acc}$ in all three scenarios).²⁴
- Although unlikely, a return to growth levels last seen several decades ago in advanced economies could still lead to improving wealth accumulation prospects for future generations, despite demographic developments.

While the above findings relate to how one year's worth of saving can potentially accumu-

²⁴Results are available but omitted for brevity.

late, I now want to explore how different generations fare once their whole life-time wealth accumulation journeys are compared. Specifically, given changing wage and return levels, how does the wealth accumulated by a worker saving throughout their working life change across generations?

Consider a worker who starts working in year t in country i and initially saves a proportion s_r of their after-tax income $(1 - \tau_w) w_{i,t}^L$, where τ_w is the tax rate on labour income. I indicate this amount as $S_{0,i,t}$, corresponding to total savings (wealth) at the end of the first year of saving when saving began at year t. This saving is invested and returns $s_r(1 - \tau_w) w_{i,t}^L \times (1 + (1 - \tau_k)r_{i,t}^k)$ at year t + 1, where τ_k is the tax rate on capital gains.²⁵ The worker adds additional savings equal to $s_r(1 - \tau_w) w_{i,t+1}^L$, such that total savings are now $S_{1,i,t} = (1 - \tau_w) w_{i,t}^L \times (1 + (1 - \tau_k)r_{i,t}^k) + (1 - \tau_w) w_{i,t+1}^L = S_{0,i,t}(1 + (1 - \tau_k)r_{i,t}^k) + (1 - \tau_w) w_{i,t+1}^L$. In general, for any $j \neq 0$:

$$S_{j,i,t} = S_{j-1,i,t} (1 + (1 - \tau_k) r_{i,t+j-1}^k) + s_r (1 - \tau_w) w_{i,t+j}^L$$
(17)

I set both τ_w and τ_k at 35%, and s_r at 20%²⁶, and I obtain an estimate of $S_{39,i,t}$, corresponding to 40 years of savings, for each country and year based on each of the three scenarios for $w^{H,nd}$. To obtain Figure 14 I go through all the steps as for Figure 13, i.e.: (i) I compute $S_{39,i,t}$; (ii) for each country, I divide it by its peak pre-2017 value across all years; (iii) for every year, I take the average of these values among the countries in the sample used for Figure 12 (averaging all the countries in the sample returns similar results).

A similar picture to Figure 13 emerges, where on average the peak in accumulated wealth is expected to be for the cohorts who started saving in the 1990s. Assuming they started working in their twenties, these agents are currently in their fifties and will likely retire starting from around year 2030. They are expected to have significantly higher accumulated wealth than the cohorts before them, but in advanced economies they will likely also be the last generation to experience a significant increase in wealth compared to their parents' generation. Depending on the forecast for $w^{H,nd}$, subsequent generations (i.e., people

 $^{^{25}}$ Taxation rates were not introduced above but being constant they have no impact in terms of the direction of any movements in W^{acc} etc. The effect of demographics through changes in tax rates is likely to make effects shown in this Section even larger than displayed here, as a larger proportion of retirees tends to lead to higher public expenditure and therefore higher tax rates, lowering after-tax savings and rates of return. To keep the exercises simple, I do not add these dynamics but it should be kept in mind as a potential further effect.

²⁶Again, the tax rate numbers are indicative and based on representative values; changing them would have an impact on the values of W^{acc} , but not its dynamics over time. As to savings rates, it would be reasonable to assume, based on the extensive literature on life-cycle consumption and savings, that savings rates are age-dependent, however here they are kept constant (with the implication that any arbitrary choice of s_r for the sake of this exercise will have no impact on the relative sizes of W^{acc} between cohorts). Nevertheless, I find that adding age-dependent savings rates to this exercise, based for instance on saving patterns like described in Feiveson and Sabelhaus (2019) (13), has only a negligible impact on results as shown below, leading to the same conclusions.

Figure 14: Wealth accumulated over 40 years' working life, as a fraction of peak value, average across selected countries.



Note: See text for computation method. The x-axis represents the year at which the agent starts saving (i.e. year t in equation 17).

currently at work and younger than 50, and their future descendants) could experience (i) progressively lower wealth accumulation (most likely scenario for advanced economies with slow growth); (ii) broadly same level of wealth accumulation as their elders; or (iii) continuously rising wealth (most likely for economies furthest from the technological frontier). Based on this, material improvements in standards of living from generation to generation might not continue to be the case in advanced economies, while to an extent it is already so for some countries.²⁷

This is not all, however. A higher level of wealth at retirement does not guarantee higher standards of living when retired, for two main reasons:

- Lower returns on capital result in lower levels of income generated from any given level of accumulated wealth.
- Longer life expectancies mean that retirees need to make sure their wealth can sustain their livelihoods for a longer period of time, decreasing the pace at which they are able to decumulate it.

Because of these facts, even if current levels of wealth accumulation were to remain, in

²⁷Southern European countries being the most striking example in this case, based on these data. See below.

several countries future generations will likely not be able to maintain the same standard of living at retirement as their parents. In this last exercise, I estimate how much retirees can expect to consume out of their accumulated wealth by investing it and decumulating it, based on the rate of return on capital, their residual life expectancy, and their initial accumulated wealth.

Consider an agent aged 65 who retires at year t in country i. They started working and saving 40 years before, and have accumulated wealth equal to $S_{i,39,t-40}$. They have an expected residual life expectancy at 65 of x but they want to provision for a higher number of years in case they live longer, say 1.25x. They want to generate a constant cash flow out of their wealth via a combination of capital returns (at rate $r_{i,t}^k$) and sale of assets, such that at year t + 1.25x they have completely decumulated all of their wealth. This situation is equivalent to a hypothetical "reverse mortgage" where the agent loans out $S_{i,39,t-40}$ for 1.25x years at $r_{i,t}^k$; every year they receive a constant cash flow payment which includes principal repayments (i.e. the amount of asset sales) plus interest (i.e. returns on the remaining capital). Based on the typical formula for mortgage amortisation, the constant annual repayments they can expect to receive are equal to:

$$A_{i,t} = S_{i,39,t-40} \left(r_{i,t}^k + \frac{r_{i,t}^k}{(1+r_{i,t}^k)^{1.25x} - 1} \right)$$
(18)

As previously, I estimate this for every country and year, and for each country I compute the annual value relative to the pre-2017 country peak, then average this for a subset of countries, as done above for Figure 14.²⁸ Figure 15 shows the results of this exercise, which are striking. See further Figure 16 for countries in Southern Europe, which are affected particularly hard by low productivity growth since 2000, and are seeing living standard prospects of retirees possibly reducing by a half by the end of the century.

On average, agents currently retiring²⁹ with the wealth accumulated during their working lives can expect to enjoy some of the highest standards of living at retirement across all cohorts in the sample; for all generations currently at work and far from retirement, prospects are expected to get progressively worse. Even under the most optimistic scenario, standards for retirees are set to decline from current levels until around 2050, and under the first and second scenarios they are seen declining all the way to the end of the projection horizon, with different intensities.

²⁸Annual residual life expectancies at 65 for the purposes of this exercise are backward induced from the UN World Population Prospects projections. This method matches existing data (where available) with a sufficient level of accuracy, but may differ somewhat.

²⁹I.e., they would have been born in the late fifties and sixties.

Figure 15: Annual cash flow from reverse mortgage of wealth accumulated over 40 years' working life, as a fraction of peak value, average across selected countries.



Note: Country-year values for the cash flow were calculated as per equation 18; the lines displayed in the figure are calculated as for Figure 13 and Figure 14. The x-axis represents the year at which the agent retires.

7 Conclusions

The findings in this paper highlight stark inter-generational inequalities in ageing economies when it comes to wealth and its accumulation, with the cohorts born at the start of the demographic transition benefiting most from demographic dynamics and growing economies, while subsequent generations are set to lose. This may be part of the reason why younger generations in many advanced economies have been expressing increasing unease at their future prospects when comparing themselves with their parents' generations.

They may be trapped in this "inverted-pyramid scheme" which is the economy of an ageing society. Normal pyramid schemes can only be sustained as long as an exponentially increasing number of participants join the scheme; once this is no longer the case, the scheme fails. While, like in a pyramid scheme, economies in modern times had been relying on an increasing number of new "entrants", which ensured rapidly increasing standards of living and wealth for all economic agents, positive demographic trends have now reversed in several economies, dragging down future prospects for generations to come. While in a pyramid scheme, those already in it benefit from new entrants (and every cohort but the last is set to gain), in a world of inverted population pyramids, new generations in the economy are Figure 16: Annual cash flow from reverse mortgage of wealth accumulated over 40 years' working life, as a fraction of peak value.



Note: Country-year values for the cash flow were calculated as per equation 18. The x-axis represents the year at which the agent retires.

increasingly penalised by the existence and size of the previous generations.

The implications of the results of this paper are clear: future generations in advanced economies can expect, under reasonable assumptions, to either accumulate less wealth (restricting their retirement income), or to save a significantly higher proportion of their labour income during their working age to avoid this (thus limiting consumption during their working age). In addition, depending on the social security system in place, they may either expect to pay higher taxes during their working life, or to receive lower pension income from the state when retired. In any case, it appears that they can reasonably expect to be worse off than previous generations as far as wealth accumulation is concerned.

While the changes in society and demographics that underpin these implications are in all likelihood here to stay, there are a number of policy actions that can help mitigate the adverse effects of the demographic dynamics we are facing.

 A return to strong technological growth, of the type experienced during the second half of the twentieth century (but more sustainable), would ensure that standards of living can remain constant or keep growing, even while populations age and decline. This would require large public and private investment in research and development and human capital which might however exacerbate the existing issues arising from abundance of capital.

- Increasing focus on productive vs unproductive assets. The increasing prices of residential capital, while other forms of capital have been depreciating, probably suggest that there is over-investment in housing. This is also partly because of a lack of available opportunities for productive investments (see point above), but it goes both ways.
- Improvements in the levels of financial education of the population. Many households lack the financial education to confidently invest in riskier asset classes, and these households are usually also less wealthy to begin with. A better financial education of all households would allow a greater proportion of them to knowingly invest in riskier but higher-yielding asset types, which have typically been held by the wealthiest in society. It would also likely channel investment into more productive forms of capital.
- Increasing retirement ages, and especially allowing agents to work longer work lives as increased life expectancies mean they can also expect to stay healthy and productive for longer.

Several complex aspects and interactions between demographic developments and the macroeconomy, which might add further dynamics and insights into the effects through which population ageing affects wealth inequality, could not be covered within the scope of this paper, but warrant further study. These include (i) the impact of inter-generational transfers on the transmission of wealth inequality from generation to generation, as agents live longer and have fewer children; (ii) dynamics related to inequality *within* generations and how this translates into inter-generational inequality; (iii) portfolio allocations and returns across different asset types, and how this affects how agents with different portfolio allocations accumulate wealth; (iv) the impact of housing on reducing or increasing wealth inequalities. Many of these issues would be best analysed within the context of a general-equilibrium overlapping-generation model and are generally not suited for an empirical approach. This is left for future research.

References

Aksoy, Yunus, Henrique S. Basso, Ron P. Smith, and Tobias Grasl (2019), "Demographic Structure and Macroeconomic Trends." American Economic Journal: Macroeconomics, 11 (1), pp. 193-222

Aiyar, Shekhar, Christian Ebeke, and Xiaobo Shao (2016), âThe impact of workforce aging on European productivity,â IMF Working Paper n. 2016/238

Antunes, Antonio, and Valerio Ercolani (2020), "Intergenerational Wealth Inequality: The Role of Demographics," Banco de Portugal Working Papers, No. 2020(9)

Börsch-Supan, Axel, Florian Heiss, Alexander Ludwig, and Joachim Winter (2003), "Pension Reform, Capital Markets and the Rate of Return," German Economic Review, Vol. 4 No. 2, pp. 151-181

Carvalho, Carlos, Andrea Ferrero, and Fernanda Nechio (2016), "Demographics and real interest rates: Inspecting the mechanism," European Economic Review, Vol. 88, September 2016, pp. 208-226

Cooper, Ian Anthony, Piet M.F.A. Sercu, and Rosanne Vanpee (2013), "The Equity Home Bias Puzzle: A Survey", Foundations and Trends in Finance, Vol. 7, No. 4, 2013

Cooley, Thomas F., and Espen Henriksen (2018), "The Demographic Deficit," Journal of Monetary Economics, Vol.93, pp. 45-62, January 2018

David, Joel M., Espen Henriksen, and Ina Simonovska (2014), âThe risky capital of emerging markets,â NBER Working Paper No. 20769, December 2014

Eichengreen, Barry (2015), "Secular Stagnation: The Long View," NBER Working Paper No. 20836, January 2015

Fair, Ray C., and Kathryn M. Dominguez (1991), âEffects of the changing U.S. age distribution on macroeconomic equations,â The Americal economic review, pp. 1276-1294

Ferrero, Giuseppe, Marco Gross, and Stefano Neri (2017), "On secular stagnation and low interest rates: Demography matters," European Central Bank Working Paper Series, No. 2088, July 2017

Feenstra, Robert C., Robert Inklaar, and Marcel P. Timmer (2015), "The Next Generation of the Penn World Table" American Economic Review, 105(10), 3150-3182, available for download at www.ggdc.net/pwt

Feiveson, Laura and John Sabelhaus (2019), "Lifecycle Patterns of Saving and Wealth Accumulation," FEDS Working Paper No. 2019-10 Geppert, Christian (2015), "On the Distributional Implications of Demographic Change," BeitrÀge zur Jahrestagung des Vereins fÃŒr Socialpolitik 2015: Ãkonomische Entwicklung - Theorie und Politik - Session: Labor 4, No. D17-V1, ZBW

Geppert, Christian, Alexander Ludwig, and Raphael Abiry (2016), "Secular Stagnation? Growth, Asset Returns and Welfare in the Next Decades: First Results," Max Planck Institute for Social Law and Social Policy Discussion Paper No. 05-2016

Goodhart, Charles A.E., and Manoj Pradhan (2017), "Demographics will Reverse Three Multi-Decade Global Trends," BIS Working Paper No. 656, August 2017

Gourinchas, Pierre-Olivier, and Jonathan A. Parker (2002), âConsumption over the life cycle,â Econometrica, Vol. 70, No. 1, pp 47-89, January 2002

Greenwald, Daniel L., Matteo Leombroni, Hanno Lustig, and Stijn Van Nieuwerburgh (2021), "Financial and Total Wealth Inequality with Declining Interest Rates," NBER Working Paper No. 28613, March 2021

Hood, Andrew, and Robert Joyce (2013), "The economic circumstances of cohorts born between the 1940s and the 1970s," Institute for Fiscal Studies Report, No. R89, 2013

Ikeda, Daisuke, and Masashi Saito (2012), "The Effects of Demographic Changes on the Real Interest Rate in Japan," Bank of Japan Working Paper Series, No. 12-E-3, February 2012

JordÃ_i, Ãscar, Katharina Knoll, Dmitry Kuvshinov, Moritz Schularick, and Alan M. Taylor (2017), âThe Rate of Return on Everything, 1870-2015â Federal Reserve Bank of San Francisco Working Paper No. 2017-25.

Kopecky, Joseph (2021), "Okay Boomer... Excess Money Growth, Inflation, and Population Aging," Trinity Economic Papers Working Paper, No. 07-21, October 2021

Kopecky, Joseph and Alan M. Taylor (2020), "The Murder-Suicide of the Rentier: Population Aging and the Risk Premium," NBER Working Paper, No. 26943, April 2020

Krueger, Dirk, Alexander Ludwig (2006), âOn the consequences of demographic change for rates of returns to capital, and the distribution of wealth and welfareâ NBER Working Paper, No. 12453

Mian, Atif R., Ludwig Straub, and Amir Sufi (2021), "The Saving Glut of the Rich," NBER Working Paper No. 26941, February 2021

Mishra, Anil V. (2015), "Measures of equity home bias puzzle," Journal of Empirical Finance, Vol. 34, pp. 293-312, 2015

Papetti, Andrea (2019), "Demographics and the natural real interest rate: historical and projected paths for the euro area," European Central Bank Working Paper Series, No. 2258, March 2019

Sajedi, Rana, and Gregory Thwaites (2016), "Why are Real Interest Rates so Low? The Role of the Relative Price of Investment Goods," IMF Economic Review, Vol. 64, No. 4, pp. 635-659

Summers, Lawrence H. (2014), "U.S. Economic Prospects: Secular Stagnation, Hysteresis, and the Zero Lower Bound," Business Economics, Vol. 49, No. 2, pp. 65-73

United Nations, Department of Economic and Social Affairs, Population Division (2019), "World Population Prospects 2019," Online Edition. Rev. 1

Vandenbroucke, Guillaume (2016), "Aging and Wealth Inequality in a Neoclassical Growth Model," Federal Reserve of St. Louis Review, Q1 2016, 98(1), pp. 61-80

A Appendix

A.1 Measures of productivity

The measure of total factor productivity used in the PWT dataset implies the assumption of a certain shape of the production function (see (12)). Though this cannot be avoided, it could potentially bias the results of any regression including productivity, which however cannot be omitted from the analysis presented in this paper for obvious economic theory reasons. In Table 5, I repeat the analysis for specification (2) of Table 1 (as this is the specification that is later used for the analysis in the other Sections) using different indicative measures of productivity that I can construct from the rest of the data. These are:

- Output per hour worked Y/H;
- Output per person employed Y/L;
- A measure of joint labour and capital productivity computed as $\frac{Y}{\sqrt{K \times H}}$, which I call $Prod_3$ here, and
- A similar measure, computed as $\frac{Y}{\sqrt{K \times L}}$, which I call $Prod_4$.

I also include TFP itself and a specification without any productivity measure. The results show several things: (i) the signs of the coefficients are not affected by the choice of productivity measure; (ii) their size, however, can change significantly; (iii) expressing productivity in terms of hours worked or per person employed does not make a difference; (iv) not having any measure of productivity leads to clear omitted variable bias. The wide range of coefficients, especially for K/N, may lead to think that, potentially, the results in the rest of the paper would not be robust to a different choice of productivity measure. This is however not the case. Replacing TFP with each of these measures, I have repeated the instrumental-variable exercise of section 4.2 and derived the combined age-group implied coefficient on $ln(r^k)$ as I have done for Figure 5 (tables omitted for brevity). The results are displayed in Figure 17 and show that the strong demographic effect on rates of return remains and is very similar regardless of the measure used. It also further supports the case for using the PWT TFP measure as the results using this measure are the closest to the average. If anything, based on the R^2 in specification (3) of Table 5, $Prod_3$ might be more appropriate; because of the larger size of the coefficients, this would lead to even stronger demographic effects on the rate of return and an even larger impact of the demographic transition on wealth accumulation as shown in Section 6.

	(1)	(2)	(3)	(4)	(5)	(6)
	$\ln(r^k)$	$\ln(r^k)$	$\ln(r^k)$	$\ln(r^k)$	$\ln(r^k)$	$\ln(r^k)$
In(α)	1.012***	1.012***	1.007***	1.007***	0.930***	0.998***
	(0.10)	(0.10)	(0.05)	(0.05)	(0.09)	(0.15)
ln(K/N)	-0.615***	-0.615***	-0.459***	-0.459***	-0.322***	-0.129***
	(0.09)	(0.09)	(0.06)	(0.06)	(0.05)	(0.05)
ln(H/L)	0.918***	0.294	0.803***	0.250	0.902***	0.232
	(0.27)	(0.24)	(0.16)	(0.17)	(0.26)	(0.30)
ln(L/N)	0.643***	0.643***	0.442***	0.442***	0.479***	0.240
	(0.14)	(0.14)	(0.07)	(0.07)	(0.13)	(0.18)
ln(Y/H)	0.624***					
	(0.11)					
ln(Y/L)		0.624***				
		(0.11)				
$In(Prod_3)$			1.105***			
			(0.06)			
$In(Prod_4)$				1.105***		
				(0.06)		
ln(TFP)					0.614***	
					(0.10)	
Constant	-2.345	-10.972***	-1.809	-9.441***	-3.878*	-1.070
	(2.12)	(2.87)	(1.70)	(1.49)	(2.09)	(2.42)
Year f.e.	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark
R^2	0.651	0.651	0.880	0.880	0.656	0.541
deg. fr.	63	63	63	63	63	63
Ν	3000	3000	3000	3000	3000	3000

Table 5: Specification (2) of Table 1, for different measures of productivity.

Figure 17: Combined effect of each age group on rates of return via ln(K/N) and ln(L/N), by measure of productivity used in instrumental variable regression (confront Tables 2 and 3).



T: +353 (0)1 224 6000 www.centralbank.ie publications@centralbank.ie

Bosca PO 559, Baile Átha Cliath 1, Éire PO Box 559, Dublin 1, Ireland